Author's response to reviews

Title: Male Reproductive Health Challenges: Appraisal of Wives' Coping Strategies

Authors:
Emmanuel Amoo (emma.amoo@covenantuniversity.edu.ng;amco50amoo@gmail.com)
Adekunbi Omideyi (akomid2000@yahoo.com)
Theophilus Fadayomi (olo_oye2005@yahoo.com)
Mofoluwake Ajayi (mofoluwake.ajayi@covenantuniversity.edu.ng)
Gbolaran Oni (gbolahan.oni@covenantuniversity.edu.ng)
Adenike Idowu (esther.idowu@covenantuniversity.edu.ng)

Version: 4 Date: 11 Oct 2016

Author’s response to reviews:


We hereby present the point-by-point response to the Editor

We considered the comments by the Editor as a serious concern and we have taken the following precise actions:

1. We solicited the service of an English major professional for the editing and general proofreading of the entire manuscript.

2. We also carefully made use of both Ginger and Grammarly software for adequate grammatical corrections including plagiarism-detection

3. We need to indicate here that several of the editing we made are beyond what can be indicated in track changes more especially because it passed through several hands.

4. However, we ensure that the editing and corrections did not affect the concise presentation of the study and understanding by prospective readers.

5. Structurally, we did not change the structure, though we interchanged research question 2 for 1 (and vice versa), we removed few redundant or unclear statements and rephrased where necessary.
6. We also did an internal review with other colleagues within the area covered by the study.

We appreciate the editor’s painstaking effort to go through the entire manuscript.